**Author’s response to reviews**

**Title:** Impact of early detection of acute invasive fungal rhinosinusitis in immunocompromised patients

**Authors:**

Mariana Silveira (dramacoelho@hotmail.com)

Wilma Anselmo-Lima (wtalima@fmrp.usp.br)

Francesca Maia (franmaia48@hotmail.com)

Danielle Queiroz (daniellecunha@ymail.com)

Rodrigo Nogueira (rodrigoln@unimedjaboticabal.coop.br)

Marcelo Leite (mjleiteorl@gmail.com)

Ricardo Lessa (dr.lessa918@gmail.com)

Belinda Simões (bpsimoes@fmrp.usp.br)

Edwin Tamashiro (edwin@fmrp.usp.br)

Fabiana Valera (facpvalera@fmrp.usp.br)

**Version:** 6  **Date:** 08 Mar 2019

**Author’s response to reviews:**

all the pending assignments were done. We hope that the article is now suitable to be considered for publication